Author’s response to reviews

Title: Efficacy of Modified Liujunzi decoction on functional dyspepsia of spleen-deficiency and qi-stagnation syndrome A randomized controlled trial

Authors:

Shengsheng Zhang (zhss2000@163.com)
Luqing Zhao (zhaoluqing111@163.com)
Hongbing Wang (ivy0106@sina.com)
Chuijie Wang (lnzyxhk@163.com)
Suiping Huang (doctorhsp@medmail.com.cn)
Hong Shen (shenhong999@163.com)
Wei Wei (weihell06666@yahoo.com.cn)
Lin Tao (gaotonglu@sohu.com)
Tao Zhou (zhoutao0903@yahoo.com)

Version: 2 Date: 23 June 2012

Author’s response to reviews: see over
Efficacy of Chinese herbal medicine on functional dyspepsia of spleen-deficiency and qi-stagnation syndrome: A randomized controlled trial

Dear editor,

Thank you for your time and efforts. According to the reviewer’s comments on our manuscript, we have finished revisions. The main amendments have been highlighted in red in the revised manuscript. We hope that the revised version is acceptable for publication in your journal.

Thank you again for your cooperation.

Sincerely yours,

Sheng-sheng Zhang
Beijing Hospital of Traditional Chinese Medicine affiliated to Capital Medical University.
E-mail: zhss2000 @163.com